



# **JGR** Earth Surface

# **RESEARCH ARTICLE**

10.1029/2020JF005762

#### **Key Points:**

- To understand how peat enters long term carbon storage requires consideration of peat accumulation over periods of 0.1–1 Myr years
- Peat accumulation over these timeframes requires sustained storage of water above depositional base level
- This has profound implications for our understanding of peat in the earth system and the interpretation of the geological record

#### Correspondence to:

D. J. Large, david.large@nottingham.ac.uk

#### **Citation**:

Large, D. J., Marshall, C., Jochmann, M., Jensen, M., Spiro, B. F., & Olaussen, S. (2021). Time, hydrologic landscape and the long-term storage of peatland carbon in sedimentary basins. *Journal* of *Geophysical Research: Earth Surface*, 126, e2020JF005762. https://doi. org/10.1029/2020JF005762

Received 30 JUN 2020 Accepted 1 FEB 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited. David J. Large<sup>1</sup>, Chris Marshall<sup>2</sup>, Malte Jochmann<sup>3,4</sup>, Maria Jensen<sup>3</sup>, Baruch F. Spiro<sup>5</sup>, and Snorre Olaussen<sup>3</sup>

<sup>1</sup>Department of and Environmental Engineering, Faculty of Engineering, University of Nottingham, University Park, Nottingham, UK, <sup>2</sup>Environmental Research Institute, University of the Highlands and Islands, North Highland College UHI, Thurso, UK, <sup>3</sup>University Centre in Svalbard (UNIS), Longyearbyen, Norway, <sup>4</sup>Store Norske Spitsbergen Kulkompani AS, Longyearbyen, Norway, <sup>5</sup>Department of Mineralogy, Natural History Museum, London, UK

**Abstract** Peatland carbon may enter long-term storage in sedimentary basins preserved as either coal or lignite. The time required to account for the carbon in 1-10 m thick coal seams must represent  $10^5$ - $10^6$  years, an order of magnitude more than previously assumed. To understand the process by which this happens requires extrapolation of our understanding of peatland carbon accumulation over timescales that greatly exceed those of Holocene peat. We analyze the consequences of extrapolating peat growth to periods of  $10^6$  years. We deduce that that key to sustained peat growth are hydrologic landscapes that can maintain a saturated peat body above the level of clastic deposition. Contrary to current stratigraphic frameworks, we conclude that the generation of accommodation space at low rates of 0.1-0.2 mm/yr can adequately accommodate thick peat accumulation over periods  $>10^5$  years. However, generation of accommodation space at rates >0.5 mm/yr cannot. The low rates that permit accommodation of thick peat are typical of the rates of subsidence in specific tectonic settings, particularly foreland basins, and this has implications for our understanding of the links between terrestrial carbon burial, tectonics and the carbon cycle. The long-term stability of extensive peatland required to form coal also requires sediment bypass, modifying basin wide sediment transport and deposition. Limits to peatland growth under very low accommodation rates must exist but the relative importance of the limiting process is not understood. Finally, we discuss the consequences of these factors for predicting the future of the peatland carbon reservoir.

**Plain Language Summary** Over the past 300 million years peatland carbon has accumulated in sedimentary rocks where it occurs as coal and lignite. This process is important as it removes carbon dioxide from the atmosphere and stores it as carbon in the Earth's crust, providing a long-term cooling mechanism for global climate. In this context, recent research has shown that layers of coal previously thought to have accumulated over 10,000 years have accumulated over periods of a hundred thousand to one million years. This marks a huge shift in our understanding of the periods over which peat can accumulate continuously. In this study, we explore the consequences of this for our understanding of the surface and tectonic processes that lead to the storage of peatland carbon in sedimentary rocks. We show that the key to continuous peat accumulation and crustal storage of carbon are landscapes that maintain stable hydrology with lower rates of subsidence than previously thought. Applying these observations, we provide a new framework for interpreting peat, lignite and coal. We then highlight links between tectonic processes, particularly the limits to peat accumulation within a landscape.

## 1. Introduction

Peatland carbon may enter into long-term crustal storage in sedimentary basins where it occurs as either coal or lignite. To understand the process by which this happens requires a conceptual bridge between peatland processes measured on Holocene timescales and processes over the much greater timescales required to explain features of substantial deposits of coal and lignite. Key to this conceptual bridge are the consequences of extrapolating Holocene peatland processes over long timeframes. However, these consequences are not considered by either peat scientists or coal geologists, and the absence of this analysis results not



only in flawed stratigraphic interpretation, but also undermines our ability to understand peat within the Earth system on timescales that greatly exceed those of the Holocene.

Coal is one of our most significant energy resources and generations of geologists have studied coal sedimentology, geochemistry and stratigraphy. Many approaches have been applied to interpret coal stratigraphy (Dai et al., 2020; C. F. K. Diessel 1992) and currently, the most favored approaches (e.g., C. Diessel et al., 2000; Holz et al., 2002; Jerrett, Davies, et al., 2011, Jerrett, Flint, et al., 2011), are founded on the stratigraphic framework proposed by Bohacs and Suter (1997). In this stratigraphic framework, growth of the water-saturated precursor peat responds to the rate of change in the space available for sediment accumulation, known as the accommodation space. The balance between the rate at which accommodation space changes (on account of tectonic subsidence, change in sea level and sediment accumulation) and the peat production rate determines the resulting thickness, areal extent and composition of the peat. This stratigraphic framework acknowledges but does not focus on other factors (e.g., climate, groundwater, vegetation, and geomorphology) that influence peat accumulation. By applying this stratigraphic framework, coal is integrated into stratigraphic models (Davies et al., 2005; Jerrett, Flint, et al., 2011; Michaelsen et al., 2000; Staub, 2002) where it forms an interpretative bridge between the marine and terrestrial realms (Wadsworth et al., 2010).

If coal is to form, the need for both peat accumulation and accommodation space is unquestionable. However, the assumptions that underpin the interpretation of coal within stratigraphic frameworks, including those of Bohacs and Suter (1997), are not firmly grounded in the processes that determine the formation and accumulation of peat. Specifically, there are two flaws. Stratigraphic frameworks consider only the volumetric growth of peat (predominantly water) and subsequent compaction, while ignoring the processes that determine carbon accumulation. They also assume that peat properties, including thickness and composition, are highly responsive to the rate at which accommodation space is generated. The first flaw results in an order of magnitude underestimate of the time required to account for the carbon in coal (Large & Marshall, 2015). The second flaw devalues the interplay between hydrology, geomorphology and water storage in determining the long-term stability and duration of a peat body. Consequently, stratigraphic interpretations based on the approach of Bohacs and Suter (1997) treat coal as a relatively transient component of the stratigraphic record, which in turn requires the inference of significant stratigraphic hiatuses (e.g., Jerrett, Flint, et al., 2011, Jerrett, Davies, et al., 2011; Scott & Stephens, 2015). It also results in coal seams that must have accumulated over periods of  $10^{5}$ - $10^{6}$  years being interpreted as would a Holocene peat with duration of  $10^4$  years, effectively ignoring the extent and potential stratigraphic and paleoenvironmental value of the record contained within the coal. Consequently, the approach also fails to recognize the capacity of peat to modify sediment transport and storage within a basin, and the capacity of coal to inform our long-term understanding of basin hydrology and its evolution relative to active structures and paleo-landscape.

The study of Holocene peat is inherently constrained by its 10 kyr timeframe and peat scientists rarely consider the consequence of extending their inferences over the timeframes in which peat may enter crustal storage or become limited within the landscape. For example, peat growth models (e.g., Baird et al., 2012; Frolking et al., 2010; Morris et al., 2012) are never extended over timeframes that greatly exceed those of the Holocene. Consequently, their validity with respect to long-term crustal storage of peat remains untested.

In this study, we define a new stratigraphic framework for the interpretation of peat, lignite and coal founded on an understanding of carbon accumulation within a hydrologic landscape. Using the carbon accumulation basis for determining time in coal, defined by Large and Marshall (2015), we determine the implications for the processes leading to crustal storage of peatland carbon over extended time scales of peat accumulation. The result is a more informed interpretative framework that encapsulates the interaction of geological structures, geomorphology and hydrology over a wide range of timescales and enables a greater understanding of peatlands within the Earth system.

# 2. Basis for a New Hydrologic Landscape Approach

In the sections that follow, we define base level as the local graded stream profile, the level at which clastic deposition and erosion balance (Catuneanu, 2006). For simplicity, and to align with the terminology used in the stratigraphic model of Bohacs and Suter (1997), the rate at which accommodation space is generated (Catuneanu, 2006) is referred to as the accommodation rate.





**Figure 1.** A van Krevelen diagram illustrating the continuous relationship between the C-O-H composition of vegetation, peat, lignite and coal. Coal data spanning the compositional range from lignite to anthracite, ranging in age from Neogene to Carboniferous is from the USGS coal quality database (n = 7,000). Mean peat, peatland and vegetation compositions are from Moore et al. (2018) and references therein. Mean peat data is shown for UK, Latvia (Lat), Ontario (Ont), Indonesia (Ind), and Southeast USA. Projection lines are indicated assuming mass is lost as mean organic matter CH<sub>2</sub>O equivalent to loss of CO<sub>2</sub> and CH<sub>4</sub> in equal proportions. Positions of average swamp and bog are shown after 60% and 70% loss of mass.

For ease of reference, we use the term coal as the generic term to refer to both coal and lignite. In this context coal is a readily combustible rock containing more than 50 percent organic matter by weight (on a dry basis), which was formed from the compaction and alteration of plant remains (Alpern & de Sousa, 2002; Jackson, 1997; Schopf, 1956). There is no single formal definition of peat. For the purpose of this paper we use the definition, modified after Schopf (1966), that peat is an un-consolidated, hydrophilic, carbonaceous sediment, formed by accumulation of partially fragmented and decomposed plant remains. The precursor to coal is always peat, however, not all materials classed as peat could form coal.

#### 2.1. Carbon Accumulation and Time

As coal is predominantly carbon, a carbon basis for determining the duration of a coal seam is far more appropriate and robust (Large & Marshall, 2015) than volumetric alternatives. The reason for this is that the processes determining the rate of carbon accumulation in peat and subsequent loss of carbon during coalification are well understood and quantifiable (Clymo, 1984; C. F. K. Diessel, 1992; Large & Marshall, 2015). For example, starting with a range of peatland vegetation chemistry (sphagnum, sedge, shrub, and wood) the relative positions of vegetation, peat and coal on a van Krevelen diagram (Figure 1) are consistent with a continuous process of transformation of vegetation to coal. Loss of CO2 and  $CH_4$  in equal proportions can account for this trend (Figure 1) as can loss of more complex mixtures including dissolved organic carbon (Moore et al., 2018). The positions of coal, peat and vegetation (Figure 1), are also consistent with a similar mix of peatland vegetation C-O-H chemistry accounting for all coal compositions throughout geological time. This illustrates that the mass balance during the transition from peat to coal is a predictable and measurable process. By accounting for the mass lost during the transition from peat to coal it is also possible to predict (Large & Marshall, 2015) the coal thickness equivalent to 20 kyr of carbon accumulation (Figure 2) for a range of carbon accumulation rates that are typical of global rates in the Holocene.

This leads to three important implications. First, the duration of coal seams is about an order of magnitude greater than previously considered for example, 1 m of bituminous coal formed under temperate paleo-climatic conditions at a carbon accumulation rate of 20 g/m<sup>2</sup>/yr would represent 100 kyr rather than the typically assumed 10 kyr (e.g., C. Diessel et al., 2000; Jerrett, Davies, et al., 2011). Second, most of the inorganic matter in coal can be accounted for by the deposition of atmospheric mineral dust at rates of 0.02–20 g/m<sup>2</sup>/yr (Large & Marshall, 2015; Marshall et al., 2016). Third, global variations in the composition of coal are predictable using uniformitarian assumptions of carbon accumulation rate and dust deposition (Large & Marshall, 2015; Marshall et al., 2016). This indicates that

First, thick coal seams (1->20 m) can be of sufficient duration to span, without hiatus, periods of 0.1->1Myr and may account for more time than inter-seam clastic sediments (Large & Marshall, 2015). This has been previously inferred from field observation. For example, Broadhurst and France (1986) and Broadhurst and Simpson (1983) noting that coal seams were more strongly influenced by active structures and associated geomorphic features than the interseam sediments concluded that the duration of coal seams was considerably greater than that of the intervening sediments. Similarly, in the UK Carboniferous, Scott







and Stephens (2015) concluded that a greater amount of time must be recorded in coal layers than by clastic sediments.

Second, coal is indicative of periods of sediment bypass during which fluvial clastic deposition is excluded. This conclusion arises as long-term siliciclastic deposition rates in coal forming environments (floodplains, estuaries and deltas) are typically in the range 0.02–1 mm per year (Einsele, 2000). At typical bulk soil densities of 1,200–1,600 kg/m<sup>3</sup> this corresponds to mass deposition rates of siliciclastic sediments of 24–1,600 g/m<sup>2</sup>/yr. Supply of siliciclastic matter to peat at rates greater than 35 g/m<sup>2</sup>/yr would result in a rock with greater than 50% inorganic matter that could not be classed as a coal, and even the lowest of these rates would produce a low-quality coal containing at least 20% inorganic matter. These long-term rates are also much less than the rates of specific clastic depositional events (Schumer & Jerolmack, 2009). This conclusion matters as there is a prevalent view that a significant proportion of the inorganic matter in some coals can originate from fluvial clastic deposition (Dai et al., 2020; C. F. K. Diessel, 1992; Glasspool, 2003)

Third, in a given paleogeographic setting, the long-term rate of carbon accumulation tends to be constant and lie within the global Holocene range. This does not mean that the rate of carbon accumulation does not vary; it only implies that given suitable hydrology the conditions of nutrient supply (via atmospheric deposition), climate and productivity will tend to produce a stable long-term average rate in a given paleogeographic setting. It is therefore more appropriate to assume that for coals of similar paleogeography and rank, thicker coal accumulated over longer periods of time, not at a faster rate.

Finally, the accumulation of peat over long periods requires sustained supply of fresh water, which in turn requires sustained hydrological stability.

## 2.2. Water Storage

With typical dry bulk densities of  $\sim 0.1$  g/cm<sup>3</sup> (Clymo, 1984), by mass and by volume, peat is predominantly water. The exclusion of clastic sediment from raised peatland in low-lying areas is therefore due to the storage of a volume of water above base level. In upland areas of blanket bog, landscape incision and consequent sediment bypass are responsible for the exclusion of clastic sediment from the peat. Therefore, accumulation and preservation of peat is not only a matter of balancing rates of peat growth and accommodation; it is also a matter of sustaining water storage in the form of peat above the level of clastic sediment deposition. This is an important and fundamental difference between this approach and previous approaches (e.g., Bohacs & Suter, 1997; C. Diessel et al., 2000).

Evidence of being either in a state of clastic supply or clastic exclusion is the typically abrupt transition from clastic sediment to coal and *vice versa* (C. F. K. Diessel, 1992). Even gradual stratigraphic transitions tend to be characterized by intercalated thin coal and clastic horizons rather than gradual dilution of clastic sediment with organic matter (C. F. K. Diessel, 1992). If we accept there is more time in the coal than the clastic sediment, then the intercalation of clastic sediment and thin coal requires extremely limited accommodation space at the point where coal seams start to establish. A conclusion that is consistent with the base of coal seams being unconformable (Haszeldine, 1989).

The hydrological stability required to initiate and sustain a volume of water above the graded stream profile is strongly influenced by the capacity of groundwater to maintain the water-saturated state of a peat body during periods of water shortage (Glaser et al., 1997, 2004). Consequently, the position of peat within a landscape is strongly related to the discharge (or seepage) of groundwater (Winter, 2000) and





**Figure 3.** A schematic representation of the fundamental hydrologic landscape unit described by Winter (2000) illustrating the optimal position for peat accumulation.

the response of this discharge to crustal deformation (Glaser et al., 2004). Peatland supported by discharge from large groundwater catchments will be resilient to climate change. Peatland associated with small groundwater catchments that are highly dependent on annual rainfall will be less resilient (Hokansen et al., 2016; Winter, 2000).

#### 2.3. The Hydrologic Landscape

Peat, as the precursor to coal, initiates in a landscape. The base of coal seams is typically unconformable, that is on land (Haszeldine, 1989). Based on analysis of controlling variables (eustasy, tectonics, climate, and vegetation), most coal seams initiate during periods of long-term lowering of global sea level (Railsback, 1995) i.e. when land is created. A useful way of conceptualizing the relationship between peat, geomorphology and hydrology on different scales, in a variety of settings, is the

hydrologic landscape concept (Winter, 2001). In this approach, Winter (2001) uses the idea of a fundamental hydrologic landscape unit. This unit consists of an upland separated from a lowland by a steeper slope (Figure 3), and incorporates both geology and climatic setting. Associated with the hydrologic landscape unit, Winter (2001), defines a hydrologic system (the components of the total hydrologic budget for any point on the landscape). This consists of surface water controlled by slope, slope aspect and surface permeability; ground water controlled by the subsurface hydraulic characteristics; atmospheric water controlled by climate. Where the supply of water is sufficient, ground water will tend to flow toward the surface (or discharge) at the base of a slope. Given a sufficiently wet climate or large enough groundwater catchment this creates the freshwater saturation required to initiate and maintain stable wetlands or peatlands (Winter, 2000). The base of slopes will therefore be the setting in which to initiate and maintain water saturated peat deposits (Figure 3).

There are two important aspects to the hydrologic landscape concept. First, the fundamental hydrologic landscape unit operates on multiple scales that can range from the break in slope between river terraces, to the transition from mountain range to basin. Scale is important because the scale of catchment determines the vulnerability and resilience of the peatland. Small catchments relating to small landscape units will be highly dependent on precipitation to sustain associated peatlands. Whereas large-scale units and associated catchments will be capable of buffering peatlands through periods of variable climate (Winter, 2000). Second, geomorphic evolution driven by tectonics and shaped by erosion evolves slowly over periods of  $10^5-10^7$  years, whereas climate and base level may display large fluctuations over much shorter timeframes. Hence, the long periods ( $10^5-10^6$  years) of hydrologic landscape stability required to account for thick coal seams (e.g., Briggs et al., 2007; Large, 2007; Large et al., 2004; Large & Marshall, 2015) must depend on hydrologic landscape units controlled by tectonics and active structures within a basin. This is essential if hydrologic stability is to be sustained during periods of higher frequency climate and base level fluctuation.

The importance of hydrologic landscape is clear in the Holocene during which long-term accumulation of peat depends more on the capacity of the landscape to sustain water storage than on the capacity to generate accommodation space. This is evidenced by the widespread accumulation of peat on various scales in a range of hydrologic landscapes (e.g., boreal lowlands, upland blanket bogs, high-altitude plateaus and coastal plains) irrespective of base level, uplift or subsidence. The short timeframe of the Holocene relative to the timescale of significant crustal deformation also tends to limit our understanding to the tectonic and structural influence of glacio-eustatic rebound on peat initiation and hydrology (Glaser et al., 2004).

An aspect not explicitly considered by Winter (2001) is that as peat accumulates it will modify the hydrologic landscape, raising the local water table and shifting the breaks in slope. An example of this phenomenon is the spring line, visible on optical satellite images, surrounding the Hongyuan peat (Large et al., 2009) and neighboring peatlands on the Qinghai-Tibetan Plateau.



#### 2.4. Influence of Changing of Base Level

A change in base level constitutes a change in the hydrologic landscape. As base level rises, surface wetness and run off increase, and new accommodation space is generated. At the level of the local graded stream profile, deposition and erosion balance, and the accommodation space is filled with clastic sediment.

In a hydrologic landscape in which peat has initiated and accumulated at the breaks in slope, rising base level is accompanied by clastic deposition away from breaks in slope. In smaller-scale hydrologic landscape units, groundwater discharge will decrease, and the water budgets required to sustain peat accumulation above base level will become increasingly dependent on local precipitation. If the water budget is insufficient to sustain a peat body above base level, the peat will be inundated by clastic sediment. As base level continues to rise, larger-scale hydrologic landscape units will maintain continuous peat.

Falling base level will reverse this trend. Water tables will fall and peat surfaces at the basin margin will become more vulnerable and collapse. However, if the scale of the hydrologic landscape units is sufficiently large, peat growth may be sustained even during periods of falling base level. At the same time, new breaks in slope associated with topographic features on the emerging and incised land surface will provide the necessary areas of groundwater seepage to initiate new peat growth.

Note that collapse of the peat surface does not necessarily mean that the peat becomes drier, as mechanically weak peat will initially collapse and track the falling water table. Only once the water table has reached the level at which the peat has sufficient strength to resist compaction will the water table continue to fall below the peat surface resulting in drying and oxidation of the peat.

Another means of modifying the relationship between base level and the hydrologic landscape is tectonic tilting. During this process, water storage will shift down the hydraulic gradient. Up gradient peat will collapse and possibly oxidize as water is withdrawn, while the down gradient time equivalent peat will be water saturated and tend to expand. In these circumstances time equivalent peats could have developed markedly different ecosystems. If the collapsed peat surface falls below base level, then the capacity exists to rapidly infill accommodation space above the peat surface. Collapse, due to water withdrawal, could explain the burial of large in situ tree trunks immediately above some coal seams (Broadhurst & Magraw, 1961; Falcon-Lang, 2006) and in areas adjacent to incised paleochannels (Guion, 1987).

#### 2.5. Accommodation and Peat Growth Models

For peat to enter long term crustal storage accommodation rates need to allow continuous peat accumulation over periods of  $10^5$ – $10^6$  years. To achieve this requires an appropriate balance between the accommodation rate and the rate of peat growth (Bohacs & Suter, 1997). This is complicated, as the rate of peat accumulation with respect to time is an inherently nonlinear balance between the input of organic matter and the mass of organic matter lost due to decay (Clymo, 1984). The nonlinearity of peat growth is ignored in previous stratigraphic models that assume linear peat growth rates (Bohacs & Suter, 1997; C. Diessel et al., 2000).

A simple evaluation of the consequences of this nonlinearity can be made by extending simple peat growth models (Clymo, 1992; Clymo et al., 1998) that provide realistic estimates of Holocene carbon accumulation over periods that greatly exceed 10 kyr. The shape of the peat growth curve generated by these models is governed by the initial decay rate, a (yr<sup>-1</sup>), the decay rule that determines the change in decay rate as decay progresses, and the rate of input of carbon to the peat, p (kmolC m<sup>-2</sup> yr<sup>-1</sup>). Reasonable estimates can be made for rate of input and initial decay rates for the Holocene, however the decay rule over long periods is unknown. We also ignore short-term ecohydrological feedback. This is not unreasonable, as over long periods the mass balance between input and decay must shape the peat growth curve. Also, the longer-term effects of ecohydrological feedback, limited by the climate space in which the peat forms, should be represented as oscillations within the long term trend.

To illustrate the influence of nonlinear peat growth we choose the quadratic model of Clymo et al. (1998) which is expressed in Equation 1:





**Figure 4.** (a) The relationship between cumulative peat thickness, cumulative peat carbon, time and linear accommodation rates (dashed lines with values given in mm/yr). Peat growth curves are derived using the quadratic model of Clymo et al. (1998). Values of input (p) and initial decay rate (a) are chosen to represent the current range of global values: extreme low latitude, low latitude, boreal and high latitude. Conversion of cumulative carbon to thickness is based on a peat dry bulk density of 0.1 g/cm<sup>3</sup> and a carbon concentration of 50% on a dry ash free basis. (b) Derived from A this illustrates the height of peat accumulation above base level for accommodation rates of 0.1 mm/yr (thin lines) and 0.2 mm/yr (thick lines). Values of p and a are those given in A. (c) The results in are expressed as bituminous coal thickness. Coal thickness was calculated from the cumulative carbon using the empirical method of Large and Marshall (2015) to account for loss of organic matter during coalification.

$$M_t = \left(\frac{p}{a}\right) \left( \left(\sqrt{1+2at}\right) - 1 \right) \tag{1}$$

where  $M_t$  is the cumulative dry mass of carbon per unit area after time t (yr). This model is only a fixed point on a continuum of possible decay rules (Clymo, 1984) and is chosen for illustrative purposes as it generates reasonable accumulation over 10<sup>4</sup> years and much longer timeframes. Some other commonly used points on the spectrum of possible decay models have unreasonable stratigraphic consequences over longer periods of time. For example, using Holocene parameters and applying a constant decay rate model could not account for any significant coal accumulation. Four growth curves were generated (Figure 4a) that cover a range of inputs and decay values. Values were chosen to represent a range of environmental conditions: high latitude (p = 0.001,  $a = 10^{-6}$ ); boreal (p = 0.003,  $a = 10^{-4}$ , similar to Clymo et al., 1998); low latitude  $(p = 0.008, a = 10^{-3}, \text{ much greater than the boreal values of Clymo$ et al., 1998); and extreme low latitude (p = 0.1, a = 0.1) equivalent to a tropical net primary production and a high decay rate in tropical soil (e.g., Yule & Gomez, 2009).

The peat growth curves (Figure 4) are then analyzed relative to linear accommodation rates, in effect constant subsidence. In reality accommodation rate may also be nonlinear, but this simplifying assumption helps focus our analyses on the consequences of nonlinear peat growth. To provide an understanding of what might be observed in terms of long-term crustal storage equivalent bituminous coal thicknesses (Figure 4c) are calculated based on the cumulative carbon and accounting for the loss of carbon during coalification (Large & Marshall, 2015).

Analysis of the relationship between accommodation rate and the modeled growth curves (Figure 4) leads to the following implications for a system in which clastic supply is sufficient to fill the available accommodation space.

Peat accumulation above base level occurs if the value of *p* and its volumetric equivalent exceeds the accommodation rate (Figures 4a and 4b). Significant peat accommodation sufficient to generate >50 cm of bituminous coal occurs at accommodation rates <0.5 mm/yr (Figure 4c). Contrary to the conclusion of Bohacs and Suter (1997) it is improbable that high accommodation rates of 1-2 mm/yr could permit thick peat accumulation. Tropical peats with high values of *p* can grow and establish above base level more rapidly (Figure 4b) than boreal peat. Over short timeframes of <70 kyr apparent long-term accumulation rates appear higher in the tropics and lower at high latitude, whereas over longer time frames this relationship reverses (Figure 4a). Whether or not that reversal is apparent over the timeframe of the Holocene is unclear, however although peat thickness is highly variable, greater Holocene peat thicknesses are more commonly reported from tropical regions (e.g., Anderson, 1983; Page et al., 1999). At accommodation rates of between 0.1 and 0.2 mm per year, which are at the extreme low end of the range used by Bohacs and Suter (1997), 100-400 kyr of continuous peat accumulation can be accommodated with the height of peat above base level not exceeding 8 m (Figure 4b). Temperate peatlands offer the maximum capacity for long-term carbon storage as they have the right balance of input



and output to enable establishment and long-term growth above base level (Figures 4a and 4b). In contrast to the conclusion of Bohacs and Suter (1997) the ratio of the accommodation rate to the peat production rate varies throughout the period of peat accumulation and bares no necessary relationship to the condition of the peatland (Figure 4a). Given sufficient time and a constant accommodation rate all peat deposits will be terminated by inundation, however, if the accommodation rate is too low, or zero, the model effectively permits indefinite peat accumulation above base level and this process must in some way be limited by erosion (Figure 4b).

## 2.6. Limits to Peat Accumulation

Stratigraphic models assume that erosion and/or unfavorable hydrology rapidly limit peat growth in areas with no accommodation or falling base level. Over the short term, such as the period of the Holocene, the influence of geomorphology overcomes this and explains why extensive peat continues to accumulate in actively eroding landscapes with no increase in accommodation space. Furthermore, reported Holocene peat thickness, for example those in the database of Loisel et al. (2014), rarely exceed a thickness of 8 m, so the influence of erosion on thicker accumulations above base level (Figure 4b) remains untested over the timeframe of the Holocene. We therefore deduce that 10 kyr is generally insufficient time to observe significant limits to peat growth. So, what processes ultimately limit peat accumulation in settings with little or no accommodation space on timescales much greater than 10 kyr? The answer to this question is significant, as considerable quantities of Holocene peat occur in uplands or areas of active post-glacial uplift with limited or no capacity for long-term crustal storage. The following two key processes limit peat growth.

## 2.6.1. Hydrological and Oxidative Limits

Without accommodation the supply of water ultimately limits peat growth in any landscape setting. As peat growth approaches the hydrological limits of the system, oxidative decay above the water table increasingly limits carbon accumulation. This intrinsic limit in which the balance between productivity and decay determines an upper limit to carbon accumulation (Clymo, 1984) has been used to predict future long-term limits (>10<sup>4</sup> years) to northern peatland carbon stocks (Alexandrov et al., 2020).

This is problematic for several reasons. By necessity it assumes constant values for hydraulic conductivity, ignoring the poro-elastic response of the multiphase (solid, liquid, gas) peat matrix. For example, peat dry bulk density in the saturated zone can vary from 0.02 to 0.2 g/cm<sup>3</sup> (Large et al., 2009; Page et al., 2004) and peat hydraulic conductivity may vary by several orders of magnitude (Charman, 2002). These illustrate that mechanical change in the pore structure of the peat can buffer the hydrology and create large uncertainty in model outputs. It ignores the role of the hydrologic landscape. Assuming some level of sustained input of atmospheric mineral dust at even quite low rates, the long-term consequences of oxidative limitation would be that the peat soil would transition into a mineral soil. For example, recent rates of dust deposition over peatland in western Siberia range from 1 to 12 g/m<sup>2</sup>/yr (Fiałkiewicz-Kozieł, 2016), ~5%–50% of the Holocene boreal long-term carbon accumulation rate (Loisel et al., 2014) before an oxidative limit has been reached. Some of this mass will be lost as soluble elements; however, most of the mass in the form of low solubility alumino-silicates, is retained (Large & Marshall, 2015).

#### 2.6.2. Mechanical Deformation and Erosion

The stability of a peat body in the absence of accommodation space must be mechanically limited. The weak peat body must eventually undergo structural failure (mass movement), drainage, and subsequent erosion, even under sustained ideal conditions for peat formation. The inevitability of such mass movement has been noted and observed in specific blanket bog settings (Tallis, 1985), however, it is not generally considered as a limiting process in all settings in which accommodation space is limited or unavailable. The surface of the peat body may also wrinkle, as proposed by Pearsall (1956) and observed in one mechanical simulation (Briggs et al., 2007), possibly into the characteristic patterns of hummocks and hollows observed in the field (Morris et al., 2013; Pearsall, 1956) and in some coal seams (Broadhurst & Simpson, 1983; Rippon, 1998). Such wrinkles could conceivably generate water channels leading to erosion, drainage and oxidation of a raised peatland under sustained wet conditions.



The accumulation and growth of peat above base level is in-effect an increase in elevation, which from the perspective of erosion is no different to tectonic uplift. Observations and theories of landscape formation (e.g., Kirby & Whipple, 2012) require that an increase in elevation of the land must be constrained by erosion, particularly so in the typically wet climates associated with peat formation. Fluvial erosion has been studied in peatlands (Gradzinski et al., 2003; Li et al., 2017; Watters et al., 2007) but often from a short-term perspective of land management and carbon budgets (Cowley et al., 2018; Evans et al., 2006; Li et al., 2017) rather than the perspective of long-term limits. This lack of consideration of the limits imposed by natural erosion probably results from two peatland characteristics. First, the fibrous nature of peat is relatively resistant to erosion creating a distinctive channel morphology and limiting channel movement (Gradzinski et al., 2003; Watters et al., 2007). Second, it is difficult to quantify natural processes of erosion in highly modified (e.g., from the effects of over grazing and trampling) and managed landscapes. An additional and notable contribution to peat erosion, particularly in upland areas is aeolian deflation, which can rapidly occur following fluvial incision (Foulds & Warburton, 2007). An ideal location in which to test the relative influence of erosion, mass movement and hydrology is the Falkland Islands. Although intensively grazed, a significant proportion of Falkland Island blanket peat initiated pre-Holocene (Payne et al., 2019), is highly dissected and in optical images and in the field shows ample evidence of established erosion and mass movement that must predate human occupation of the islands.

## 2.7. Hiatuses Within Peat and Coal

In many interpretations, the lateral equivalent of stratigraphic boundaries generated on account of base-level fluctuations within a subsiding basin are interpreted as intra-seam hiatuses in coal and vice versa. Evidenced by change in humification without notable siliciclastic deposition these hiatuses are interpreted to represent periods on the order of  $10^4$ – $10^6$  years (Davies et al., 2005; Holdgate et al., 1995; Jerrett, Davies, et al., 2011). This is improbable. The generation of depositional breaks in peat requires sustained erosion or nonaccumulation of the peat surface without accommodating clastic input or atmospheric dust. This is difficult, as peat will continue to lose mass due to on-going decay, in effect creating its own accommodation space. Furthermore, organic rich peat surfaces are naturally attractive sites for plant growth and, even in the absence of clastic deposition re-establishment of peat accumulation is likely on periods much less than  $10^4$  years. Evidence to support this are the short periods (about  $10^2$ – $10^3$  years) that account for hiatuses in Holocene peat known as recurrence surfaces (Borgmark, 2005). The short duration of these time gaps is not surprising as even the loss of the top 1 m of a typical Holocene peatland need not result in a stratigraphic gap of more than  $10^3$  years (Borgmark, 2005; Page et al., 2004). Consequently, in a subsiding basin, it is improbable that intra-seam discontinuities without clastic deposition could represent long periods.

#### 2.8. Principles of the New Hydrologic Landscape Approach to the Crustal Storage of Peat

Based on the above it is possible to define a set of considerations and related guidelines for the interpretation of peat and coal.

First and foremost, the hydrologic landscape (Figure 3) determines the position, volume and extent of a peat body. Within that landscape thick peat forms in areas of prolonged hydrological stability the resilience of which is determined by the geometry, scale and persistence of the hydrological landscape unit. Relative change in base level will modify the hydrologic landscape but does not preclude the capacity of the hydrologic landscape to maintain peat growth above base level.

With respect to coal, in a given paleogeographic and paleoclimate setting, thicker coal seams accumulated over longer periods, not at a different rate, and the thickness of a coal seam is proportional to the time integrated carbon accumulation rate. The formation of coal requires clastic sediment bypass and hence storage of water in the precursor peat above the level of the equilibrium stream profile. Therefore, changes into and out of periods of coal formation is a response to the change in the volume of water stored above base level.

#### 3. Discussion

Previously considered transient, coal records are now demonstrably of sufficient duration to capture global change in a manner analogous to an ice core. This most significant change in interpretation arises from the order of magnitude increase in time required to account for the carbon in coal. The capacity to ascribe



timescales to coal records without worrying about independent chemostratigraphic or biostratigraphic tiepoints enables a huge post-Devonian archive of global environmental change contained in coal to be re-evaluated. This archive has the potential to yield time calibrated records of atmospheric deposition (Marshall et al., 2016), fire (Belcher et al, 2003; Scott & Glasspool, 2006), palynology (D. J. Nichols & Warwick, 2005; Phillips et al, 1985), organic geochemistry (Bechtel et al., 2008; Benner et al., 1987), plant and atmospheric  $\delta^{13}$ C (Arens et al., 2000), paleotemperature (Inglis et al., 2017), and much more. Indeed, given the potential duration of coal seams it is unsurprizing rather than fortuitous that several major global events have been reported from within coal seams (Collinson et al, 2009; Steinthorsdottir et al., 2016; Wang et al., 2018) and a more appropriate internal timeframe should greatly assist their interpretation.

Given sufficiently thick coal, predicted timeframes may also be tested directly. For example using Ar-Ar radiometric ages from associated volcanic ash deposits (Chen et al., 2014), or Re-Os dates from marine-influenced coal (Tripathy et al., 2015). Uncertainty in the radiogenic age is often greater than the likely duration of the seam and this may constrain useful application. Timeframes may also be indirectly tested relative to stratigraphic tie-points and whether or not they remove or create stratigraphic anomalies. For example, when our estimates of time in coal are applied to the Duckmantian-Langsettian interval of the Carboniferous, the time accounted for in the clastic sediments and coal increases from 75 to 125 kyr (Scott and Stephens, 2015) to 0.7–1.5 Myr (Large & Marshall, 2015). The former value requires huge unsubstantiated hiatuses whilst the latter does not and still lies within chronostratigraphic estimates for the duration of the interval.

Our approach has the capacity to enhance understanding of syn-depositional basin processes and the tectonic environments in which peat may enter crustal storage. Our postulated rates of basin subsidence required to sustain peat accumulation are encountered in many tectonic settings (Allen & Allen, 2013) associated with significant accumulations of coal. On extensional continental margins, such as the coal-bearing Gippsland Basin (Birch, 2003), subsidence rates range from 0.2 mm/yr at the initiation of rifting to less than 0.05 mm/yr during the flexural subsidence phase, over time scales of 10<sup>6</sup>–10<sup>7</sup> years (Allen & Allen, 2013). Foreland basins subside at rates of 0.2–0.5 mm/yr, and cratonic basins at 0.01–0.04 mm/yr (Allen & Allen, 2013). Coals in the cratonic basins of Australia are notably oxidized, a property that is a logical consequence of particularly low rates of subsidence (Hunt & Smyth, 1989).

Foreland basins, in particular, hold vast coal deposits. For example, the Cretaceous western interior and Laramide forelands in North America (Dickinson et al., 1988; Pederson & Dehler, 2005), the Gondwana Panthalassa foreland (Veevers & Powell, 1994), and the late Carboniferous Variscan foreland in Europe (Mazur et al., 2010; Suess et al., 2007). They also collect rainwater and are first order hydrological basins (Garcia-Castellanos, 2002) or, indeed, first order hydrological landscapes. So, given a suitable climate, does the structural and hydrological evolution of a foreland basin favor the stable hydrologic landscapes required to sustain peat accumulation? If such a logical link exists on timeframes of  $10^5$ – $10^7$  years then the concepts presented in this paper should enable a more informed incorporation of terrestrial carbon accumulation into models that integrate tectonics, hydrology, geomorphic evolution and the carbon cycle (e.g., Wong et al., 2019).

Evidence of such a link between the tectonic cycle and peat entering crustal storage may be the coincidence between type III kerogen burial (predominantly coal), an increase in the global ratio of the rates of carbon to sulfur burial (Berner, 2003), and the burial of extensive carbon in the Variscan, Gondwana (Veevers, 2013), and Cretaceous western interior (Molenaar and Rice 1988) foreland basins. In this context, Berner (2003) notes that a prominent peak in the ratio of carbon to sulfur burial corresponds to a lower-than-expected proportion of type III kerogen burial (coal) in the Late Carboniferous and Permian. A possible explanation for this anomaly is that the rate of Paleozoic type III kerogen burial, determined from the volumetric proportion of coal in sedimentary formations, is markedly underestimated relative to its Mesozoic and Cenozoic equivalents and not directly comparable to C/S burial derived from syndepositional stable isotope values. Peat, up to 20–40 Myr post-deposition, will typically be lignite. Carboniferous and Permian peat, ~300 Myr post deposition is typically high rank bituminous coal or anthracite. The volume of Carboniferous coal will therefore be almost half that of its Cenozoic lignite equivalent (Figure 2). Syndepositional rates of type III kerogen deposition almost twice those estimated from current coal deposits, easily remove the late Carboniferous and Permian deficit in type III kerogen burial.



The hydrological landscape requirements for sustained accumulation of peat mean that coal seam thickness should be a sensitive indicator of syn-depositional geomorphic evolution within a basin. Using our approach in combination with isopach maps, variations in coal thickness can be used to map the evolution of active structures and landscape within a basin. A specific example of this is the detailed work of Marshall (2013) who demonstrated the role of persistent structures in controlling the geomorphology, distribution and thickness of early Paleocene coal within the Central Tertiary Basin, Svalbard. There are also numerous other examples (e.g., Ferm & Staub. 1985; Greb et al., 2001; Guion, 1987; Haszeldine, 1989; Read, 1989) linking coal thickness to syn-depositional topography, differential compaction, and tectonics. However, they have not been interpreted in the context of a hydrologic landscape over an appropriate timeframe and re-analysis of these data is beyond the scope of this paper.

A hydrological landscape that sustains the stability of peatland above depositional base level must also sustain sediment bypass and modify the supply, transport, and distribution of sediment within a basin. Basinward, a period of extensive peat accumulation should be marked by enhanced sediment supply and associated progradation, for example, of thick shoreface sands. Such an association is observed in many coal-bearing paralic systems. For example, extensive coastal peat deposits are observed to pass laterally into thick shoreface sediments within the Central Tertiary Basin, Spitsbergen (Bruhn & Steel, 2003; Marshall, 2013), the Cretaceous South-Western Interior Basin, USA (Pederson & Dehler, 2005), the Cretaceous Blackhawk Formation, USA (Jerrett et al., 2011a) and in the Gippsland Basin, Australia (Birch, 2003). In all cases greater significance is placed on the capacity of clastic sediments to yield insights into basin evolution and stratigraphy, yet the coal may provide a more continuous record of syndepositional processes.

A set of implications also arise with respect to Holocene peat and the projected long-term carbon sequestration capacity of modern peatlands. Whilst there is a good knowledge of decay rates over periods <10 kyr and coalification rates over periods greater than 10 Myr there is nothing between. Therefore the only certainty on which long term models of continuous peat accumulation can be based is the nonlinearity of the mass balance between inputs and outputs. Extension of peat accumulation models over much longer periods does however provide a test of whether a given model is a reasonable basis for understanding crustal carbon storage. For example, using a Holocene parameterization, the constant decay model of Clymo et al. (1998) can provide an adequate approximation to peat growth over 10 kyr but places an absolute and nonsensical limit of <30 cm on the total thickness of coal that could ever be stored over periods >10<sup>6</sup> years.

In the absence of accommodation space, there must be a limit to the quantity of peat a given landscape can hold. Therefore, an important question is how quickly, and over what spatial and temporal scales, erosion will limit the size of the peatland carbon reservoir in different geomorphic settings. Answers to this could have profound consequences for current and future management of peatland and for predictions of the capacity of the peatland carbon reservoir (Alexandrov et al., 2020; J. E. Nichols & Peteet, 2019). For example, upland blanket bog should reach its natural erosional limits faster than lowland raised bogs. Evidence of this is the eroded state of upland blanket bog (Tallis, 1985) and although land management practices may confound this judgment (Tallis, 1985), it can be argued that areas closer to their natural limits will be more vulnerable to environmental pressure. The inevitability of an erosional limit in landscapes where accommodation space is not being created also requires that we account for both spatial and temporal erosional losses when attributing a carbon flux to large areas of peatland. It also implies that in landscapes without the capacity to accommodate long-term accumulation of peat, the long-term management strategy should be to minimize rates of carbon loss rather than promote metastable growth above the natural limits imposed by erosion. In terms of global carbon budgets we can question where tectonics and hydrologic landscape are likely to enhance or limit the longevity of peat accumulation. Areas with the potential for long-term peat accumulation leading to crustal storage include the Sunda Shelf and Peruvian Amazon Basin, both foreland basins (DeCelles & Giles, 1996; Rasanen et al., 1992) holding large quantities of peat (Householder et al., 2012; Xu et al., 2018). Whereas, peatlands of the cratonic and isostatically uplifting Hudson Bay Lowlands and Scandinavia seem unlikely prospects for crustal storage.

## 4. Conclusions

To account for the crustal carbon contained within peat, coal, and lignite requires considerably longer periods than previously considered. In turn, this requires consideration of the parameters of the hydrologic landscape that can sustain the conditions suitable for peat formation. This approach brings coal strati-



graphic models in line with modern peat observations and can be applied to all peatland and coal settings on various scales. It sets limits on the tectonic environments that enable long-term accumulations of peat leading to coal and demonstrates that slow crustal deformation alone can sustain the hydrologic landscape necessary for thick coal formation. The sustained periods of peat formation needed to account for the carbon in coal require recognition of a prolonged and fundamental shift in the depositional environment and basinwide processes. The hydrologic landscape approach facilitates re-interpretation of these systems and raises questions as to the ultimate limits to peat accumulation.

## **Conflicts of Interest**

The authors declare no conflicts of interest relevant to this study.

## **Data Availability Statement**

The paper does not contain new data.

## References

- Alexandrov, G. A., Brovkin, V. A., Kleinen, T., & Yu, Z. (2020). The capacity of northern peatlands for long-term carbon sequestration. *Biogeosciences*, 17(1), 47–54. https://www.biogeosciences.net/17/47/2020/
- Allen, P. A., & Allen, J. R. (2013). In Basin Analysis: Principles and Application to Petroleum Play Assessment (3rd ed. p. 632.) Wiley-Blackwell.
- Alpern, B., & de Sousa, M. J. L. (2002). Documented international enquiry on solid sedimentary fossil fuels; coal: Definitions, classifications, reserves-resources, and energy potential. *International Journal of Coal Geology*, 50(1–4), 3–41. https://doi.org/10.1016/ S0166-5162(02)00112-X
- Anderson, J. A. R. (1983). The tropical peat swamps of western Malesia. *Mires: Swamp, bog, fen and moor: Regional studies*. Elsevier Scientific Publishing Company. https://ci.nii.ac.jp/naid/10019557618/en/
- Arens, N. C., Jahren, A. H., & Amundson, R. (2000). Can C3 plants faithfully record the carbon isotopic composition of atmospheric carbon dioxide?. Paleobiology, 26, 137–164. https://doi.org/10.1666/0094-8373(2000)026<0137:CCPFRT>2.0.CO;2
- Baird, A. J., Morris, P. J., & Belyea, L. R. (2012). The DigiBog peatland development model 1: Rationale, conceptual model, and hydrological basis. *Ecohydrology*, 5(3), 242–255. https://doi.org/10.1002/eco.230
- Bechtel, A., Gratzer, R., Sachsenhofer, R. F., Gusterhuber, J., Lucke, A., & Puttmann, W. (2008). Biomarker and carbon isotope variation in coal and fossil wood of Central Europe through the Cenozoic. *Palaeogeography, Palaeoclimatology, Palaeoecology, 262*(3–4), 166–175. https://doi.org/10.1016/j.palaeo.2008.03.005
- Belcher, C. M., Collinson, M. E., Sweet, A. R., Hildebrand, A. R., & Scott, A. C. (2003). Fireball passes and nothing burns-The role of thermal radiation in the Cretaceous-Tertiary event: Evidence from the charcoal record of North America. *Geology*, 31(12), 1061–1064. https://doi.org/10.1130/G19989.1
- Berner, R. A. (2003). The long-term carbon cycle, fossil fuels and atmospheric composition. *Nature*, 426(6964), 323–326. https://doi.org/10.1038/nature02131
- Benner, R., Fogel, M. L., Sprague, E. K., & Hodson, R. E. (1987). Depletion of C-13 in lignin and its implications for stable carbon isotope studies. *Nature*, 329(6141), 708–710. https://doi.org/10.1038/329708a0

Birch, W. D. (Ed.), (2003). Geology of Victoria. Geological Society of Australia (Victoria Division).

Bohacs, K., & Suter, J. (1997). Sequence stratigraphic distribution of coaly rocks: Fundamental controls and paralic examples. Aapg Bulletin-American Association of Petroleum Geologists, 81(10), 1612–1639. https://doi.org/10.1306/3B05C3FC-172A-11D7-8645000102C1865D

- Borgmark, A. (2005). Holocene climate variability and periodicities in south-central Sweden, as interpreted from peat humification analysis. *Holocene*, *15*(3), 387–395. https://doi.org/10.1191/0959683605hl816rp
- Briggs, J., Large, D. J., Snape, C., Drage, T., Whittles, D., Cooper, M., et al. (2007). Influence of climate and hydrology on carbon in an early Miocene peatland. *Earth and Planetary Science Letters*, 253(3–4), 445–454. https://doi.org/10.1016/j.epsl.2006.11.010
- Broadhurst, F. M., & France, A. A. (1986). Time represented by coal seams in the coal measures of England. International Journal of Coal Geology, 6(1), 43–54. https://doi.org/10.1016/0166-5162(86)90024-8

Broadhurst, F. M., & Magraw, D. (1961). On a fossil tree found in an opencast coal site near Wigan, Lancashire. *Geological Journal*, 2(2), 155–158. https://doi.org/10.1002/gj.3350020203

- Broadhurst, F. M., & Simpson, I. M. (1983). Syntectonic sedimentation, rigs, and fault reactivation in the coal measures of Britain. Journal of Geology, 91(3), 330–337. https://doi.org/10.1086/628775
- Bruhn, R., & Steel, R. (2003). High-resolution sequence stratigraphy of a clastic foredeep succession (Paleocene, Spitsbergen): An example of peripheral-bulge-controlled depositional architecture. *Journal of Sedimentary Research*, 73(5), 745–755. https://doi. org/10.1306/012303730745
- Catuneanu, O. (2006). Principles of sequence stratigraphy. Elsevier Science.
- Charman, D. (2002). Peatlands and environmental change. Chichester: John Wiley and Sons.
- Chen, Z. L., Ding, Z. L., Tang, Z. H., Wang, X., & Yang, S. L. (2014). Early Eocene carbon isotope excursions: Evidence from the terrestrial coal seam in the Fushun Basin, Northeast China. *Geophysical Research Letters*, 41(10), 3559–3564. https://doi.org/10.1002/2014gl059808
  Clymo, R. S. (1984). The limits to peat bog growth. *Philosophical Transactions of the Royal Society of London Series B-Biological Sciences*, 303(1117), 605–654. https://doi.org/10.1098/rstb.1984.0002
- Clymo, R. S. (1992). Models of peat growth. Suo, 43, 127-136. https://doi.org/10.2307/3547057

Clymo, R. S., Turunen, J., & Tolonen, K. (1998). Carbon accumulation in peatland. Oikos, 81(2), 368–388. https://doi.org/10.2307/3547057

#### Acknowledgments

This work was jointly funded by Natural Environment Research Council grants NE/G523871/1 and NE/ P014100/1, the Store Norske Spitsbergen Kulkompani AS and the Faculty of Engineering, University of Nottingham. The University Centre in Svalbard is thanked for providing office space, expertize and library access during research visits. M., Waddington and R.S., Clymo are thanked for providing comments on the manuscript.



- Collinson, M. E., Steart, D. C., Harrington, G. J., Hooker, J. J., Scott, A. C., Allen, L. O., et al. (2009). Palynological evidence of vegetation dynamics in response to palaeoenvironmental change across the onset of the Paleocene-Eocene Thermal Maximum at Cobham, Southern England. *Grana*, 48(1), 38–66. https://doi.org/10.1080/00173130802707980
- Cowley, K., Looman, A., Maher, D. T., & Fryirs, K. (2018). Geomorphic controls on fluvial carbon exports and emissions from upland swamps in eastern Australia. Science of The Total Environment, 618, 765–776. https://doi.org/10.1016/j.scitotenv.2017.08.133
- Dai, S., Bechtel, A., Eble, C. F., Flores, R. M., French, D., Graham, I. T., et al. (2020). Recognition of peat depositional environments in coal: A review. *International Journal of Coal Geology*, 219, 103383. https://doi.org/10.1016/j.coal.2019.103383
- Davies, R., Diessel, C., Howell, J., Flint, S., & Boyd, R. (2005). Vertical and lateral variation in the petrography of the Upper Cretaceous Sunnyside coal of eastern Utah, USA-implications for the recognition of high-resolution accommodation changes in paralic coal seams. *International Journal of Coal Geology*, 61(1–2), 13–33. https://doi.org/10.1016/j.coal.2004.06.003
- $DeCelles, P. G., \&\ Giles, K. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin systems. Basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. A. (1996). Foreland basin Research, 8(2), 105-123. https://doi.org/10.1046/j.1365-2117.1996.01491.x. Foreland basin Research, 8(2),$
- Dickinson, W. R., Klute, M. A., Hayes, M. J., Janecke, S. U., Lundin, E. R., Mckittrick, M. A., & Olivares, M. D. (1988). Paleogeographic and Paleotectonic Setting of Laramide Sedimentary Basins in the Central Rocky-Mountain Region. *Geological Society of America Bulletin*, 100(7), 1023–1039. https://doi.org/10.1130/0016-7606(1988)100<1023:Papsol>2.3.Co;2
- Diessel, C., Boyd, R., Wadsworth, J., Leckie, D., & Chalmers, G. (2000). On balanced and unbalanced accommodation/peat accumulation ratios in the Cretaceous coals from Gates Formation, Western Canada, and their sequence-stratigraphic significance. *International Journal of Coal Geology*, 43(1–4), 143–186. https://doi.org/10.1016/S0166-5162(99)00058-0

Diessel, C. F. K. (1992). Coal bearing depositional systems. (p. 721). Berlin: Springer. https://doi.org/10.1007/978-3-642-75668-9

Einsele, G. (2000). Sedimentary basins: Evolution, facies, and sediment budget (2nd ed.). Berlin: Springer-Verlag. https://doi. org/10.1007/978-3-662-04029-4

- Evans, M., Warburton, J., & Yang, J. (2006). Eroding blanket peat catchments: Global and local implications of upland organic sediment budgets. *Geomorphology*, 79(1–2), 45–57. https://doi.org/10.1016/j.geomorph.2005.09.015
- Falcon-Lang, H. J. (2006). A history of research at the Joggins Fossil Cliffs of Nova Scotia, Canada, the world's finest Pennsylvanian section. *Proceedings of the Geologists Association*, 117, 377–392. https://doi.org/10.1016/S0016-7878(06)80044-1
- Ferm, J. C., & Staub, J. R. (1985). Depositional controls of mineable coal bodies. In R. A. Rahmani & R. M., Flores (Eds.), Sedimentology of Coal and Coal-Bearing Sequences (pp. 273–289). Wiley. https://doi.org/10.1002/9781444303797.ch15
- Fialkiewicz-Koziel, B., Smieja-Krol, B., Frontasyeva, M., Slowinski, M., Marcisz, K., Lapshina, E., et al. (2016). Anthropogenic- and natural sources of dust in peatland during the Anthropocene. *Scientific Reports*, 6(1), 38731. https://doi.org/10.1038/srep38731
- Foulds, S. A., & Warburton, J. (2007). Wind erosion of blanket peat during a short period of surface desiccation (North Pennines, Northern England). Earth Surface Processes and Landforms, 32(3), 481–488. https://doi.org/10.1002/esp.1422
- Frolking, S., Roulet, N. T., Tuittila, E., Bubier, J. L., Quillet, A., Talbot, J., & Richard, P. J. H. (2010). A new model of Holocene peatland net primary production, decomposition, water balance, and peat accumulation. *Earth System Dynamics*, 1(1), 1–21. https://doi.org/10.5194/ esd-1-1-2010
- Garcia-Castellanos, D. (2002). Interplay between lithospheric flexure and river transport in foreland basins. *Basin Research*, 14(2), 89–104. https://doi.org/10.1046/j.1365-2117.2002.00174.x
- Glaser, P. H., Siegel, D. I., Reeve, A. S., Janssens, J. A., & Janecky, D. R. (2004). Tectonic drivers for vegetation patterning and landscape evolution in the Albany River region of the Hudson Bay Lowlands. *Journal of Ecology*, 92(6), 1054–U1052. https://doi. org/10.1111/j.0022-0477.2004.00930.x
- Glaser, P. H., Siegel, D. I., Romanowicz, E. A., & Shen, Y. P. (1997). Regional linkages between raised bogs and the climate, groundwater, and landscape of north-western Minnesota. *Journal of Ecology*, 85(1), 3–16. https://doi.org/10.2307/2960623
- Glasspool, I. J. (2003). Hypautochthonous-allochthonous coal deposition in the Permian, South African, Witbank Basin No. 2 seam; a combined approach using sedimentology, coal petrology and palaeontology. *International Journal of Coal Geology*, 53(2), 81–135. https:// doi.org/10.1016/S0166-5162(02)00193-3
- Gradzinski, R., Baryla, J., Doktor, M., Gmur, D., Gradzinski, M., Kedzior, A., et al. (2003). Vegetation-controlled modern anastomosing system of the upper Narew River (NE Poland) and its sediments. *Sedimentary Geology*, 157(3–4), 253–276. https://doi.org/10.1016/ S0037-0738(02)00236-1
- Greb, S. F., Eble, C. F., Williams, D. A., & Nelson, W. J. (2001). Dips, ramps, and rolls–evidence for paleotopographic and syn-depositional fault control on the Western Kentucky No. 4 coal bed, Tradewater Formation (Bolsovian), Illinois Basin. *International Journal of Coal Geology*, 45(4), 227–246. https://doi.org/10.1016/S0166-5162(00)00023-9
- Guion, P. D. (1987). The influence of a palaeochannel on seam thickness in the coal measures of Derbyshire, England. *International Journal of Coal Geology*, 7(3), 269–299. https://doi.org/10.1016/0166-5162(87)90040-1
- Haszeldine, R. S. (1989). Coal reviewed: Depositional controls, modern analogues and ancient climates. *Geological Society, London, Special Publications*, 41(1), 289. https://doi.org/10.1144/GSL.SP.1989.041.01.20
- Hokanson, K. J., Lukenbach, M. C., Devito, K. J., Kettridge, N., Petrone, R. M., & Waddington, J. M. (2016). Groundwater connectivity controls peat burn severity in the boreal plains. *Ecohydrology*, 9, 574–584. https://doi.org/10.1002/eco.1657
- Holdgate, G. R., Kershaw, A. P., & Sluiter, I. R. K. (1995). Sequence stratigraphic analysis and the origins of Tertiary brown coal lithotypes, Latrobe Valley, Gippsland Basin, Australia. *International Journal of Coal Geology*, 28(2–4), 249–275. https://doi. org/10.1016/0166-5162(95)00020-8
- Holz, M., Kalkreuth, W., & Banerjee, I. (2002). Sequence stratigraphy of paralic coal-bearing strata: An overview. International Journal of Coal Geology, 48(3–4), 147–179. https://doi.org/10.1016/S0166-5162(01)00056-8
- Householder, J. E., Janovec, J. P., Tobler, M. W., Page, S., & Lahteenoja, O. (2012). Peatlands of the Madre de Dios River of Peru: Distribution, geomorphology, and habitat diversity. Wetlands, 32(2), 359–368. https://doi.org/10.1007/s13157-012-0271-2
- Hunt, J. W., & Smyth, M. (1989). Origin of inertinite-rich coals of Australian cratonic basins. International Journal of Coal Geology, 11(1), 23–46. https://doi.org/10.1016/0166-5162(89)90111-0
- Inglis, G. N., Collinson, M. E., Riegel, W., Wilde, V., Farnsworth, A., Lunt, D. J., et al. (2017). Mid-latitude continental temperatures through the early Eocene in western Europe. *Earth and Planetary Science Letters*, 460, 86–96. https://doi.org/10.1016/j.epsl.2016.12.009 Jackson, J. A. (1997). *Glossary of geology* (4th ed. p. 769). American Geological Institute.
- Jerrett, R. M., Davies, R. C., Hodgson, D. M., Flint, S. S., & Chiverrell, R. C. (2011a). The significance of hiatal surfaces in coal seams. Journal of the Geological Society, 168(3), 629–632. https://doi.org/10.1144/0016-76492010-178

Jerrett, R. M., Flint, S. S., Davies, R. C., & Hodgson, D. M. (2011b). Sequence stratigraphic interpretation of a Pennsylvanian (Upper Carboniferous) coal from the central Appalachian Basin, USA. *Sedimentology*, 58(5), 1180–1207. https://doi.org/10.1111/j.1365-3091.2010.01200.x



- Kirby, E., & Whipple, K. X. (2012). Expression of active tectonics in erosional landscapes. *Journal of Structural Geology*, 44, 54–75. https://doi.org/10.1016/j.jsg.2012.07.009
- Large, D. J. (2007). A 1.16 Ma record of carbon accumulation in western European peatland during the Oligocene from the Ballymoney lignite, Northern Ireland. *Journal of the Geological Society*, 164, 1233–1240. https://doi.org/10.1144/0016-76492006-148
- Large, D. J., Jones, T. F., Briggs, J., Macquaker, J. H. S., & Spiro, B. F. (2004). Orbital tuning and correlation of 1.7 My of continuous carbon storage in an early Miocene peatland. *Geology*, *32*(10), 873–876. https://doi.org/10.1130/G20824.1
- Large, D. J., & Marshall, C. (2015). Use of carbon accumulation rates to estimate the duration of coal seams and the influence of atmospheric dust deposition on coal composition. *Geological Society, London, Special Publications*, 404(1), 303–315. https://doi.org/10.1144/ sp404.15
- Large, D. J., Spiro, B., Ferrat, M., Shopland, M., Kylander, M., Gallagher, K., et al. (2009). The influence of climate, hydrology and permafrost on Holocene peat accumulation at 3500 m on the eastern Qinghai-Tibetan Plateau. *Quaternary Science Reviews*, 28(27–28), 3303–3314. https://doi.org/10.1016/j.quascirev.2009.09.006
- Li, P. F., Irvine, B., Holden, J., & Mu, X. M. (2017). Spatial variability of fluvial blanket peat erosion rates for the 21st Century modelled using PESERA-PEAT. Catena, 150, 302–316. https://doi.org/10.1016/j.catena.2016.11.025
- Loisel, J., Yu, Z. C., Beilman, D. W., Camill, P., Alm, J., Amesbury, M. J., et al. (2014). A database and synthesis of northern peatland soil properties and Holocene carbon and nitrogen accumulation. *Holocene*, 24(9), 1028–1042. https://doi.org/10.1177/0959683614538073
- Marshall, C. (2013). Palaeogeographic development and economic potential of the coal-bearing palaeocene Todalen Member, Spitsbergen. (PhD). University of Nottingham. Retrieved from http://eprints.nottingham.ac.uk/id/eprint/13794
- Marshall, C., Large, D. J., & Heavens, N. G. (2016). Coal-derived rates of atmospheric dust deposition during the Permian. Gondwana Research, 31, 20–29. https://doi.org/10.1016/j.gr.2015.10.002
- Mazur, S., Aleksandrowski, P., Turniak, K., Krzeminski, L., Mastalerz, K., Gorecka-Nowak, A., et al. (2010). Uplift and late orogenic deformation of the Central European Variscan belt as revealed by sediment provenance and structural record in the Carboniferous foreland basin of western Poland. *International Journal of Earth Sciences*, 99(1), 47–64. https://doi.org/10.1007/s00531-008-0367-3
- Michaelsen, P., Henderson, R. A., Crosdale, P. J., & Mikkelsen, S. O. (2000). Facies architecture and depositional dynamics of the Upper Permian Rangal Coal Measures, Bowen Basin, Australia. *Journal of Sedimentary Research*, 70(4), 879–895. https://doi. org/10.1306/2dc4093f-0e47-11d7-8643000102c1865d
- Molenaar, C. M., Rice, D. D., & Sloss, L. L. (1988). Cretaceous rocks of the Western Interior Basin. Sedimentary Cover—North American Craton. (Vol. D-2). Geological Society of America. https://doi.org/10.1130/DNAG-GNA-D2.77
- Moore, T. R., Large, D., Talbot, J., Wang, M., & Riley, J. L. (2018). The stoichiometry of carbon, hydrogen, and oxygen in peat. Journal of Geophysical Research: Biogeosciences, 123, 3101–3110. https://doi-org.ezproxy.nottingham.ac.uk/10.1029/2018JG004574
- Morris, P. J., Baird, A. J., & Belyea, L. R. (2012). The DigiBog peatland development model 2: Ecohydrological simulations in 2D. Ecohydrology, 5(3), 256–268. https://doi.org/10.1002/eco.229
- Morris, P. J., Baird, A. J., & Belyea, L. R. (2013). The role of hydrological transience in peatland pattern formation. *Earth Surface Dynamics*, *I*(1), 29–43. https://doi.org/10.5194/esurf-1-29-2013
- Nichols, D. J., & Warwick, P. D. (2005). Palynology in coal systems analysis—The key to floras, climate, and stratigraphy of coal-forming environments. Geological Society of America. https://doi.org/10.1130/0-8137-2387-6.51
- Nichols, J. E., & Peteet, D. M. (2019). Rapid expansion of northern peatlands and doubled estimate of carbon storage. *Nature Geoscience*, 12(11), 917. https://doi.org/10.1038/s41561-019-0454-z

Page, S. E., Rieley, J. O., Shotyk, W., & Weiss, D. (1999). Interdependence of peat and vegetation in a tropical peat swamp forest. *Philosophi-cal transactions of the Royal Society of London. Series B, Biological Sciences*, 354(1391), 1885–1897. https://doi.org/10.1098/rstb.1999.0529

- Page, S. E., Wust, R. A. J., Weiss, D., Rieley, J. O., Shotyk, W., & Limin, S. H. (2004). A record of Late Pleistocene and Holocene carbon accumulation and climate change from an equatorial peat bog (Kalimantan, Indonesia): Implications for past, present and future carbon dynamics. *Journal of Quaternary Science*, 19(7), 625–635. https://doi.org/10.1002/jqs.884
- Payne, R. J., Ring-Hrubesh, F., Rush, G., Sloan, T. J., Evans, C. D., & Mauquoy, D. (2019). Peatland initiation and carbon accumulation in the Falkland Islands. *Quaternary Science Reviews*, 212, 213–218. https://doi.org/10.1016/j.quascirev.2019.03.022

Pearsall, W. (1956). Two blanket-bogs in Sutherland. Journal of Ecology, 44(2), 493-516. https://doi.org/10.2307/2256833

- Pederson, J. L., & Dehler, C. M. (2005). Interior western United States. Geological society of America field guide. Geological Society of America. https://doi.org/10.1130/0-8137-0006-X
- Phillips, T. L., Peppers, R. A., & Dimichele, W. A. (1985). Stratigraphic and interregional changes in Pennsylvanian Coal-Swamp Vegetation–Environmental Inferences. *International Journal of Coal Geology*, 5(1–2), 43. https://doi.org/10.1016/0166-5162(85)90010-2
- Railsback, L. B. (1995). Controls on long-term global rates of coal deposition, and the link between Eustasy and global geochemistry. In B. U. Haq (Ed.), Sequence Stratigraphy and Depositional Response to Eustatic, Tectonic and Climatic Forcing. Coastal Systems and Continental Margins. Vol. 1. Dordrecht: Springer. https://doi.org/10.1007/978-94-015-8583-5
- Rasanen, M., Neller, R., Salo, J., & Jungner, H. (1992). Recent and ancient fluvial deposition systems in the Amazonian Foreland Basin, Peru. *Geological Magazine*, 129(3), 293–306. https://doi.org/10.1017/S0016756800019233
- Read, W. A. (1989). The influence of basin subsidence and depositional environment on regional patterns of coal thickness within the Namurian fluvio-deltaic sedimentary fill of the Kincardine Basin. Scotland. *Geological Society, London, Special Publications, 41*(1), 333. https://doi.org/10.1144/GSL.SP.1989.041.01.23
- Rippon, J. H. (1998). The identification of syn-depositionally-active structures in the coal-bearing Upper Carboniferous of Great Britain. *Proceedings of the Yorkshire Geological Society*, 52(1), 73–93. https://doi.org/10.1144/pygs.52.1.73
- Schopf, J. M. (1956). A definition of coal. Economic Geology, 51(6), 521–527. https://doi.org/10.2113/gsecongeo.51.6.521
- Schopf, J. M. (1966). Definitions of peat and coal and of graphite that terminates the coal series (Graphocite). *The Journal of Geology*, 74(5), 584–592.
- Schumer, R., & Jerolmack, D. J. (2009). Real and apparent changes in sediment deposition rates through time. Journal of Geophysical Research, 114. https://doi.org/10.1029/2009jf001266
- Scott, A. C., & Glasspool, I. J. (2006). The diversification of Paleozoic fire systems and fluctuations in atmospheric oxygen concentration. Proceedings of the National Academy of Sciences of the United States of America, 103(29), 10861–10865. https://doi.org/10.1073/ pnas.0604090103
- Scott, A. C., & Stephens, R. S. (2015). British Pennsylvanian (Carboniferous) coal-bearing sequences: Where is the time?. Geological Society, London, Special Publications, 404(1), 283–302. https://doi.org/10.1144/sp404.14
- Staub, J. R. (2002). Marine flooding events and coal bed sequence architecture in southern West Virginia. International Journal of Coal Geology, 49(2–3), 123–145. https://doi.org/10.1016/S0166-5162(01)00074-X



- Steinthorsdottir, M., Vajda, V., & Pole, M. (2016). Global trends of pCO(2) across the Cretaceous-Paleogene boundary supported by the first southern Hemisphere stomatal proxy-based pCO(2) reconstruction. Palaeogeography, Palaeoclimatology, Palaeoecology, 464, 143–152. https://doi.org/10.1016/j.palaeo.2016.04.033
- Suess, M. P., Drozdzewski, G., & Schaefer, A. (2007). Sedimentary environment dynamics and the fort-nation of coal in the Pennsylvanian Variscan foreland in the Ruhr Basin (Germany, Western Europe). *International Journal of Coal Geology*, 69(4), 267–287. https://doi. org/10.1016/j.coal.2006.05.003
- Tallis, J. H. (1985). Mass movement and Erosion of a southern Pennine Blanket Peat. Journal of Ecology, 73(1), 283-315. https://doi.org/10.2307/2259784
- Tripathy, G. R., Hannah, J. L., Stein, H. J., Geboy, N. J., & Ruppert, L. F. (2015). Radiometric dating of marine-influenced coal using Re-Os geochronology. Earth and Planetary Science Letters, 432, 13–23. https://doi.org/10.1016/j.epsl.2015.09.030
- Veevers, J. J. (2013). Pangea: Geochronological correlation of successive environmental and strati-tectonic phases in Europe and Australia. Earth-Science Reviews, 127, 48–95. https://doi.org/10.1016/j.earscirev.2013.09.001
- Veevers, J. J., & Powell, C. M. (1994). Permian-Triassic pangean basins and foldbelts along the panthalassan margin of gondwanaland. Boulder, CO: Geological Society of America. https://doi.org/10.1130/MEM184
- Wadsworth, J., Diessel, C., & Boyd, R. (2010). The sequence stratigraphic significance of paralic coal and its use as an indicator of accommodation space in terrestrial sediments. Application of Modern Stratigraphic Techniques: Theory and Case Histories, 94, 201–219. https:// doi.org/10.2110/sepmsp.094.201
- Wang, J., Shao, L-Y., Spiro, B., & Large, D. (2018). SHRIMP zircon U–Pb ages from coal beds across the Permian–Triassic boundary, eastern Yunnan, southwestern China. Journal of Palaeogeography, 7, 117–129. https://doi.org/10.1016/j.jop.2018.01.002
- Watters, J. R., & Stanley, E. H. (2007). Stream channels in peatlands: The role of biological processes in controlling channel form. Geomorphology, 89(1–2), 97–110. https://doi.org/10.1016/j.geomorph.2006.07.015
- Winter, T. C. (2000). The vulnerability of wetlands to climate change: A hydrologic landscape perspective. Journal of the American Water Resources Association, 36(2), 305–311. https://doi.org/10.1111/j.1752-1688.2000.tb04269.x
- Winter, T. C. (2001). The concept of hydrologic landscapes. Journal of the American Water Resources Association, 37(2), 335–349. https://doi.org/10.1111/j.1752-1688.2001.tb00973.x
- Wong, K., Mason, E., Brune, S., East, M., Edmonds, M., & Zahirovic, S. (2019). Deep carbon cycling over the past 200 million years: A review of fluxes in different tectonic settings. *Frontiers in Earth Science*, 7, 263. https://doi.org/10.3389/feart.2019.00263
- Xu, J. R., Morris, P. J., Liu, J. G., & Holden, J. (2018). PEATMAP: Refining estimates of global peatland distribution based on a meta-analysis. Catena, 160, 134–140. https://doi.org/10.1016/j.catena.2017.09.010
- Yule, C. M., & Gomez, L. N. (2009). Leaf litter decomposition in a tropical peat swamp forest in Peninsular Malaysia. Wetlands Ecology and Management, 17, 231–241. https://doi.org/10.1007/s11273-008-9103-9